Author’s response to reviews

Title: Negative Pressure Wound Therapy versus modified Barker Vacuum Pack as temporary abdominal closure technique for open abdomen management: a four-year experience.

Authors:

Giulia Montori (giulia.montori@gmail.com)
Niccolo Allievi (niccolo.allievi@gmail.com)
Federico Coccolini (federico.coccolini@gmail.com)
Leonardo Solaini (leondosolaini@googlemail.com)
Luca Campanati (lcampanati@asst-pg23.it)
Marco Ceresoli (marco.ceresoli89@gmail.com)
Paola Fugazzola (paola.fugazzola@gmail.com)
Roberto Manfredi (rmanfredi@asst-pg23.it)
Stefano Magnone (smagnone@asst-pg23.it)
Matteo Tomasoni (matteotomasoni83@gmail.com)
Luca Ansaloni (lansaloni@asst-pg23.it)

Version: 4 Date: 06 Jul 2017

Author’s response to reviews:

Thank you very much for accepting the current article with minor revision and thank you for your input.

The abstract was updated as requested; emails for all the Authors were presented in the first page. The name of the ethical approval is now in the Manuscript. The declaration section was amended as necessary.

All the highlighted text was eliminated and a clean copy updated.